Author’s response to reviews

Title: Retroperitoneoscopic radical nephrectomy with a small incision for renal cell carcinoma: Comparison with the conventional method

Authors:

Hiroki Ito (pug_daikichi@yahoo.co.jp)
Kazuhide Makiyama (makiya@yokohama-cu.ac.jp)
Takashi Kawahara (takashi tk2001@yahoo.co.jp)
Futoshi Sano (fsano@yokohama-cu.ac.jp)
Takayuki Murakami (t-murazo@wb3.so-net.ne.jp)
Narihiko Hayashi (twnary@yahoo.co.jp)
Yasuhide Miyoshi (miyoyasu@med.yokohama-cu.ac.jp)
Noboru Nakaigawa (nakaigan@med.yokohama-cu.ac.jp)
Masahiro Yao (masayao@med.yokohama-cu.ac.jp)
Yoshinobu Kubota (urol_sec@yokohama-cu.ac.jp)

Version: 3 Date: 1 August 2011

Author’s response to reviews: see over
Dear Editors:

Enclosed please find our manuscript entitled, “Retroperitoneoscopic radical nephrectomy with a small incision for renal cell carcinoma: Comparison with the standard method” revised version, which I am submitting for publication in Journal of negative results in biomedicine as a original adult article.

This manuscript was the revised version.

We corrected this manuscript according to pointing out in major revisions sections.

Sincerely yours,
Hiroki Ito

Authors: Hiroki Ito, Kazuhide Makiyama, Takashi Kawahara, Takayuki Murakami, Yasuhide Miyoshi, Noboru Nakaigawa, Masahiro Yao, Yoshinobu Kubota

Institution: Department of Urology, Yokohama City University Graduate School of Medicine and School of Medicine

Postal address: 3-9, Fukuura, Kanazawa-ku, Yokohama-City, Kanagawa, Japan

Telephone number and fax number: 81-044-572-2035

E-mail address: pug_daikichi@yahoo.co.jp